Author's response to reviews

**Title:** Relationships between obesity, glycemic control, and cardiovascular risk factors: a pooled analysis of cross-sectional data from Spanish patients with type 2 diabetes in the preinsulin stage

**Authors:**

Luis A Vázquez (vazquez_salvi_luis_alberto@lilly.com)
Ángel Rodríguez (rodriguez_angel@lilly.com)
Javier Salvador (jsalvador@unav.es)
Juan F Ascaso (ascaso@uv.es)
Helmut Petto (petto_helmut@lilly.com)
Jesús Reviriego (reviriego_jesus@lilly.com)

**Version:** 3  
**Date:** 6 May 2014

**Author's response to reviews:**

Line and page numbers have been added to the manuscript as requested.

The names of the ethics committees that approved each of the observational studies that are analysed in this data mining manuscript are as follows:

H8O-MC-B003 - Comité Ético de Investigación Clínica del Hospital Universitario la Paz
H8O-XM-B011 - Comité Ético de Investigación Clínica del Hospital Universitario la Paz
F3Z-XM-B003 – Comité Ético de Investigación Clínica del Hospital Clínico San Carlos
H6E-XM-S001 - Comité Ético de Investigación Clínica del Hospital Universitario Vall d’Hebron
H6E-XM-B002 - Comité Ético de Investigación Clínica del Hospital Clínico San Carlos

Please note that we have not added this information to the manuscript as we did not feel that it was necessary. We hope this is okay with you.